Author's response to reviews

Title: Infiltrating Ductal Carcinoma Breast with Central Necrosis Closely Mimicking Ductal Carcinoma in Situ (Comedo Type): A Potential Serious Diagnostic Error

Authors:

Shahid Pervez (shahid.pervez@aku.edu)
Hassan Khan (drhasankhan@gmail.com)

Version: 2 Date: 2 May 2007

Author's response to reviews:

We have made the requested changes in accordance with the author guidelines. If there is still anything required we will be willing to provide it.

Regards

Professor Dr Shahid Pervez